Author's response to reviews

Title: Temporal trends and gender differentials in causes of childhood deaths at Ballabgarh, India - Need for revisiting child survival strategies

Authors:

Anand Krishnan Dr. (kanandiyer@yahoo.com)
Nawi Ng Dr. (nawi.ng@epiph.umu.se)
Suresh Kumar Kapoor Dr. (drskkapoor@gmail.com)
Chandrakant Sambhaji Pandav Dr. (cpandav@gmail.com)
Peter Byass Dr. (peter.byass@epiph.umu.se)

Version: 4 Date: 9 July 2012

Author's response to reviews:

Dear Editor,

The re-revised paper is being submitted. The results and conclusions sections of abstract has been revised as per reviewers suggestion. Comment on lack of data to assess causality between private sector and child mortality decline has been added to discussion.

Paper in a track change mode has also been uploaded.